A psychological intervention by community pharmacies to prevent depression in adults with subthreshold depression and long-term conditions: the CHEMIST pilot RCT

Elizabeth Littlewood,^{1*} Carolyn A Chew-Graham,² Elizabeth Coleman,¹ Samantha Gascoyne,¹ Claire Sloan,¹ Shehzad Ali,^{1,3} Jay Badenhorst,⁴ Della Bailey,¹ Suzanne Crosland,¹ Charlotte EW Kitchen,¹ Dean McMillan,^{1,5} Caroline Pearson,¹ Adam Todd,^{6,7} Cate Whittlesea,⁸ Clare Bambra,⁶ Catherine Hewitt,¹ Claire Jones,⁹ Ada Keding,¹ Elizabeth Newbronner,¹ Alastair Paterson,¹⁰ Shelley Rhodes,¹¹ Eloise Ryde,^{1,12} Paul Toner,^{1,13} Michelle Watson,¹ Simon Gilbody^{1,5} and David Ekers^{1,12}

¹Department of Health Sciences, University of York, York, UK

²School of Primary, Community and Social Care, Keele University, Keele, UK

³Department of Epidemiology and Biostatistics, Western University, London, ON, Canada

⁴Whitworth Chemists Ltd, Foxhills Industrial Estate, Scunthorpe, UK

⁵Hull York Medical School, University of York, York, UK

⁶Institute of Population Health Sciences, Newcastle University, Newcastle upon Tyne, UK ⁷School of Pharmacy, Newcastle upon Tyne, UK

⁸University College London School of Pharmacy, University College London, London, UK

⁹Public Health Team, Adult & Health Services, Durham County Council, Durham, UK

¹⁰Pharmacy Department, Northumberland, Tyne and Wear NHS Foundation Trust, Newcastle upon Tyne, UK

¹¹University of Exeter Medical School, University of Exeter, Exeter, UK

¹²Research and Development, Tees, Esk and Wear Valleys NHS Foundation Trust, Middlesbrough, UK

¹³Centre for Improving Health-Related Quality of Life, School of Psychology, Queen's University Belfast, Belfast, UK

^{*}Corresponding author liz.littlewood@york.ac.uk

Declared competing interests of authors: Jan Badenhorst reports being the Superintendent Pharmacist for Whitworth Chemists Ltd (Newcastle upon Tyne, UK), of which some pharmacies took part in the Community Pharmacy Mood Intervention Study (CHEMIST), and is chairperson of Tees Local Pharmaceutical Committee representing community pharmacies in Teesside, where some pharmacies involved in CHEMIST were based. Adam Todd is currently a member of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Prioritisation Committee A (2020–present). This committee aims to develop funding calls in the areas of primary and social care. Catherine Hewitt reports being a member of the NIHR HTA Commissioning Sub Board (2016–17) and NIHR HTA Commissioning Committee (2015–20). Simon Gilbody has been a member of several NIHR Committees (2008–20). David Ekers was a member of the National Institute for Health and Care Excellence Depression Guideline development group and is a member of the NIHR HTA programme funding committee. Carolyn Chew-Graham reports personal fees from West Midlands National Institute for Health Research Applied Research Collaboration during the conduct of the study.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published March 2022 DOI: 10.3310/EKZE0617

Plain English summary

The CHEMIST pilot RCT

Public Health Research 2022; Vol. 10: No. 5

DOI: 10.3310/EKZE0617

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

People with long-term physical health conditions (e.g. diabetes) often experience a few symptoms of depression (e.g. 'low mood'), but these symptoms often go untreated in mental health services. Community pharmacies may be a good place to offer mental health support because they are easy to access by most people. We wanted to adapt a psychological treatment ('behavioural activation') aimed at reducing low-mood symptoms in people with long-term conditions to make this suitable for delivery by pharmacy staff. Pharmacy staff and general practices identified suitable people ('participants') and we trained pharmacy staff ('support workers') to deliver the treatment.

In the first phase, we tested the study procedures and asked participants and support workers about the treatment to help us determine what changes were required. After refinements, we ran a small pilot study and compared participants who were offered the treatment with participants who continued with their usual care. All participants were asked to complete questionnaires when they started the study and again after 4 months.

Fewer people than we needed took part in the pilot study. However, the majority of participants engaged with the study and completed questionnaires at 4 months. Most people in the treatment arm started treatment and most of these completed two or more of the six sessions offered, with some participants reporting improvements in mood and behaviour. Participants and support workers told us that they felt that the treatment was understandable and that the community pharmacy was a suitable place to deliver this treatment. Pharmacy staff reported finding it difficult to incorporate the study and deliver the treatment within their busy workloads.

The study findings suggest that pharmacy staff can be trained to deliver this treatment but that more research is needed to overcome the problems with identifying participants and incorporating the delivery of psychological treatments into pharmacy routines.

Public Health Research

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health Research (NIHR), is the leading UK funder of public health research, evaluating public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health.

For more information about the PHR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/public-health-research.htm

This report

The research reported in this issue of the journal was funded by the PHR programme as project number 14/186/11. The contractual start date was in January 2017. The final report began editorial review in February 2020 and was accepted for publication in October 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2022. This work was produced by Littlewood *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk